Author's response to reviews

Title: Neuroendocrine Tumors of the Gallbladder: Review of the Literature and 44th case report.

Authors:

Silvia Mezi (Silvia.mezi@uniroma1.it)

Version: 2 Date: 4 March 2011

Author's response to reviews:

MS: 3305780284603753
Neuroendocrine tumors of the gallbladder: Review of the literature and 44th case report.
Silvia Mezi

1) A full section on gallbladder polyps management has been included in discussion (four relative references added), as required by Pavan Chepyala.

2) Language has been review and language corrections made.

3) The correct number of patients analysed in SEER between 1973-2005 has been reported, and the value has been corrected in the test.

4) Histological findings has been reported not in separate view, as initially done, because the case was not considered coherently reported by both Reviewers.

5) Patient follow-up has been up to dated.

6) All changes from the previous version are underlines.

With best wishes
Silvia Mezi